import TypeList from "@site/src/components/TypeList"

# services

## Classes

- [AnalyticsConfigService](../services/classes/services.AnalyticsConfigService.mdx)
- [AuthService](../services/classes/services.AuthService.mdx)
- [BatchJobService](../services/classes/services.BatchJobService.mdx)
- [CartService](../services/classes/services.CartService.mdx)
- [ClaimItemService](../services/classes/services.ClaimItemService.mdx)
- [ClaimService](../services/classes/services.ClaimService.mdx)
- [CurrencyService](../services/classes/services.CurrencyService.mdx)
- [CustomShippingOptionService](../services/classes/services.CustomShippingOptionService.mdx)
- [CustomerGroupService](../services/classes/services.CustomerGroupService.mdx)
- [CustomerService](../services/classes/services.CustomerService.mdx)
- [DiscountConditionService](../services/classes/services.DiscountConditionService.mdx)
- [DiscountService](../services/classes/services.DiscountService.mdx)
- [DraftOrderService](../services/classes/services.DraftOrderService.mdx)
- [EventBusService](../services/classes/services.EventBusService.mdx)
- [FulfillmentProviderService](../services/classes/services.FulfillmentProviderService.mdx)
- [FulfillmentService](../services/classes/services.FulfillmentService.mdx)
- [GiftCardService](../services/classes/services.GiftCardService.mdx)
- [IdempotencyKeyService](../services/classes/services.IdempotencyKeyService.mdx)
- [LineItemAdjustmentService](../services/classes/services.LineItemAdjustmentService.mdx)
- [LineItemService](../services/classes/services.LineItemService.mdx)
- [MiddlewareService](../services/classes/services.MiddlewareService.mdx)
- [NewTotalsService](../services/classes/services.NewTotalsService.mdx)
- [NoteService](../services/classes/services.NoteService.mdx)
- [NotificationService](../services/classes/services.NotificationService.mdx)
- [OauthService](../services/classes/services.OauthService.mdx)
- [OrderEditItemChangeService](../services/classes/services.OrderEditItemChangeService.mdx)
- [OrderEditService](../services/classes/services.OrderEditService.mdx)
- [OrderService](../services/classes/services.OrderService.mdx)
- [PaymentCollectionService](../services/classes/services.PaymentCollectionService.mdx)
- [PaymentProviderService](../services/classes/services.PaymentProviderService.mdx)
- [PaymentService](../services/classes/services.PaymentService.mdx)
- [PriceListService](../services/classes/services.PriceListService.mdx)
- [PricingService](../services/classes/services.PricingService.mdx)
- [ProductCategoryService](../services/classes/services.ProductCategoryService.mdx)
- [ProductCollectionService](../services/classes/services.ProductCollectionService.mdx)
- [ProductTypeService](../services/classes/services.ProductTypeService.mdx)
- [ProductVariantInventoryService](../services/classes/services.ProductVariantInventoryService.mdx)
- [ProductVariantService](../services/classes/services.ProductVariantService.mdx)
- [ProductService](../services/classes/services.ProductService.mdx)
- [RegionService](../services/classes/services.RegionService.mdx)
- [ReturnReasonService](../services/classes/services.ReturnReasonService.mdx)
- [ReturnService](../services/classes/services.ReturnService.mdx)
- [SalesChannelInventoryService](../services/classes/services.SalesChannelInventoryService.mdx)
- [SalesChannelLocationService](../services/classes/services.SalesChannelLocationService.mdx)
- [SalesChannelService](../services/classes/services.SalesChannelService.mdx)
- [SearchService](../services/classes/services.SearchService.mdx)
- [ShippingOptionService](../services/classes/services.ShippingOptionService.mdx)
- [ShippingProfileService](../services/classes/services.ShippingProfileService.mdx)
- [StagedJobService](../services/classes/services.StagedJobService.mdx)
- [StoreService](../services/classes/services.StoreService.mdx)
- [StrategyResolverService](../services/classes/services.StrategyResolverService.mdx)
- [SwapService](../services/classes/services.SwapService.mdx)
- [SystemPaymentProviderService](../services/classes/services.SystemPaymentProviderService.mdx)
- [TaxProviderService](../services/classes/services.TaxProviderService.mdx)
- [TaxRateService](../services/classes/services.TaxRateService.mdx)
- [TokenService](../services/classes/services.TokenService.mdx)
- [TotalsService](../services/classes/services.TotalsService.mdx)
- [UserService](../services/classes/services.UserService.mdx)
